Author's response to reviews

Title: Weight Science: evaluating the evidence for a paradigm shift

Authors:

    Linda Bacon (linda@lindabacon.org)
    Lucy Aphramor (aa0059@coventry.ac.uk)

Version: 4 Date: 17 December 2010

Author's response to reviews: see over
To the Editors at Nutrition Journal

Thank you for sending the reviewers’ comments. We believe this feedback has strengthened the paper and are pleased to re submit our revised article with changes highlighted as requested. We also attach a separate sheet in response to reviewers’ comments.

One of the authors has previously published with Nutrition Journal and so we are aware of its advantages in terms of the fast turn around time from acceptance to posting on the journal site. We would be grateful if we could be informed of the date when the article will be published in Nutrition Journal in advance, and would appreciate if publication could be delayed until 2011. This will enable us to maximise our publicity.

Thank you for your consideration. We look forward to hearing from you.

Yours sincerely,

Linda Bacon, PhD
Lucy Aphramor, PhD Cand, RD